Author's response to reviews

Title: Protection of p53 wild type cells from taxol by nutlin-3 in the combined lung cancer treatment

Authors:
- Sergey V Tokalov (Sergey.Tokalov@oncoray.de)
- Nasreddin D Abolmaali (Nasreddin.Abolmaali@oncoray.de)

Version: 4 Date: 19 February 2010

Author's response to reviews: see over
Dear Sirs,

Many thanks for your kind letter of 19 February regarding the manuscript: Tokalov SV, Abolmaali ND “Protection of p53 wild type cells from taxol by nutlin-3 in the combined treatment of NSCLC”.

For your convenience, I shall go through the points raised in the revisions (comments are indicated by bold), explain our response (blue) and include changes in the respective parts of the text (indicated in bold red).

Sincerely yours,

Dr Sergey Tokalov
Authors' contributions: Please sue this section to describe the role of each author in the study. Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. Please state that all authors read and approved the final manuscript.

According to requires next sentences were included in the manuscript. The statement was performed.

SVT carried out the literature search, designed the study performed in vitro cell culture, cell treatment, flow cytometry, statistical analysis, and drafted the manuscript. NDA coordinated the study and drafted the manuscript.